Author’s response to reviews

Title: Depression care management for late-life depression in China primary care: protocol for a randomized controlled trial

Authors:

Shulin Chen (csl@zju.edu.cn)
Yeates Conwell (yeates_conwell@urmc.rochester.edu)
Baihua Xu (xubaihua501@126.com)
Helen Chiu (helenchiu@cuhk.edu.hk)
Xin Tu (xin_tu@urmc.rochester.edu)
Yan Ma (ma_yan666@21cn.com)

Version: 3 Date: 22 March 2011

Author’s response to reviews: see over
Dear Editor,

This is a manuscript by Shulin Chen entitled “Depression care management for late-life depression in China primary care: protocol for a randomized controlled trial”. It is submitted to be considered for publication as an “Original Research” in your journal. This paper is new. Neither the entire paper nor any part of its content has been published or has been accepted elsewhere. It is not being submitted to any other journal. All authors have seen the manuscript and approved to submit to your journal. We have not entered into an agreement with a funding organization that limited our ability to complete the research as planned and to publish the results. We have had full control of all primary data. Furthermore, we have read and have abided by the statement of ethical standards for manuscripts submitted to Trials.

Correspondence should be addressed to Shulin Chen at the following address, phone and fax number, and email address:

Shulin Chen
Department of Psychology, Zhejiang University, China
Address: 148 Tianmushan Road, Xixi Campus of Zhejiang University, Hangzhou, Zhejiang, China 310028
Email: csl@zju.edu.cn Tel: +86-571-88273336 Fax: +86-571-88273337

Thanks very much for your attention to our paper.

Sincerely,

Shulin Chen
3/15/2011